\begin{table}%t1
\caption{\label{sandint1}A list of fiber-fed IFUs and their respective data-reduction pipelines.}
%\centering
\par
\begin{tabular}{ll@{\hspace{10pt}}l@{\hspace{10pt}}cclc}
\hline\hline \noalign{\smallskip}
Telescope & \multicolumn{1}{c}{Spectrograph} & IFU & $n_{\rm d}$ & Ref. & Reduction tool/Pipeline & Ref.\\
\hline
VLT/UT2 & GIRAFFE & FLAMES-ARGUS   & 1 & 1 & \textsc{bldrs} & 1a\\
        &         &                &   &   & \textsc{giraffe pipeline} & 1b\\
Gemini North/South && GMOS-N, GMOS-S &3& 2 &                &   \\
Magellan I         && IMACS          &8& 3 &                & 3a\\
WHT                &WYFFOS& INTEGRAL       &1& 4 \\
Calar Alto 3.5 m    &PMAS &LARR      &1& 5 & \textsc{p}3\textsc{d}, \textsc{p}3\textsc{d\_online}  & 5a\\
                   &     &PPAK      &1& 6 & \textsc{ppak\_online}\\
AAT & AAOMEGA      & SPIRAL         &2& 7 & 2\textsc{dfdr} & 7a\\
VLT/UT3 &VIMOS     & VIMOS-IFU      &4& 8 & \textsc{vipgi} & 8a\\
      &&             &&& \textsc{vimos pipeline}\\
McDonald 2.7 m      &VIRUS-P& VIRUS-P     &1   &  9 & \textsc{vaccine} &9a\\[1.0ex]
\hline
\end{tabular}
\tablefoot {In Col.~4 we specify the number of detectors of the IFU. Column~5 specifies the main instrument reference paper, and Col.~7 gives the reference of each instrument-specific reduction tool/pipeline.}\vspace{-2mm}
\tablebib {$^1$ \citet{AvGuJo.:03}; $^{1{\rm a}}$ \citet{BlCaNo.:00}; $^{1{\rm b}}$ \citet{PaAvAl.:00}; $^2$ \citet{AlMuCo.:02}; $^3$ \citet{ScDoCo.:04}; $^{3{\rm a}}$~\citet{BoBu:07}; $^4$ \citet{ArCaCa.:98}; $^5$ \citet{RoKeFe.:05}; $^{5{\rm a}}$ \citet{TBe:02}; $^6$ \citet{KeVeRo.:06}; $^7$ \citet{SmSaBr.:04}; $^{7{\rm a}}$ \citet{ShSaSm.:06}; $^{8}$~\citet{LeSaMa.:03}; $^{8{\rm a}}$ \citet{ScFrGa.:05}; $^{9}$ \citet{HiMaSm.:08}; $^{9{\rm a}}$ Adam et~al., in prep.}
\end{table}